Author’s response to reviews

Title: Is the association between acne and mental distress influenced by diet? Results from a cross-sectional population study among 3775 late adolescents in Oslo, Norway.

Authors:
Jon A Halvorsen (j.a.halvorsen@medisin.uio.no)
Florence Dalgard (fdalgard@ibcc.harvard.edu)
Magne Thoresen (magne.thoresen@medisin.uio.no)
Espen Bjertness (espen.bjertness@medisin.uio.no)
Lars Lien (lars.lien@medisin.uio.no)

Version: 4 Date: 11 August 2009

Author’s response to reviews:

Dear editor.

I have now gone through the manuscript once more and corrected some small typographical errors. In addition, figure 1 is uploaded separately.

Best regards

Jon Anders Halvorsen